Author's response to reviews

Title: A preliminary retrospective cohort study into the treatment of Retinopathy of Prematurity with topical ketorolac

Authors:

Medardo M Avila-Vazquez (medarav@mater.fcm.unc.edu.ar)
Roque R Maffrand (rmaffrand@uolsinectis.com.ar)
Mirta M Sosa (mirtasosa@hotmail.com)
Maria MA Franco (mariaangped@hotmail.com)
Beatriz MB Vaca de Alvarez (vacamb@onenet.com.ar)
Maria ML Cafferata (cafferaml@adinet.com.uy)

Version: 5 Date: 20 Apr 2004

shortening the manuscript
modifying the title
to insist with the preliminary characteristic of the study
highlighting the limitations of the study
correction and revision of the translation to English